Author's response to reviews

Title: Community health workers can improve child growth of antenatally-depressed, South African mothers: A cluster randomized controlled trial

Authors:

Mark Tomlinson Prof (markt@sun.ac.za)
Mary Jane Rotheram-Borus Prof (MOtheram@mednet.ucla.edu)
Jessica Harwood Dr (JHarwood@mednet.ucla.edu)
Ingrid Le Roux Dr (ingridleroux@gmail.com)
Mary O'Connor Dr (MOConnor@mednet.ucla.edu)
Carol Worthman Dr (worthman@emory.edu)

Version: 7 Date: 14 August 2015

Author's response to reviews: see over
Dear Editor

**Paper:** Community health workers can improve child growth of antenatally-depressed, South African mothers: A cluster randomized controlled trial  
**Authors:** Mark Tomlinson, Mary Jane Rotheram-Borus, Jessica Harwood, Ingrid le Roux, Mary O’Connor and Carol Worthman

We have made the following revisions to the manuscript:

1. We have formatted the Abstract into Background, Methods, Results and Conclusions sections.
2. We have included Keywords after the Abstract.
3. We have included the tables at the end of the main manuscript.
4. We have included Abbreviations and Competing Interest sections after the Conclusions section.
5. We have removed details of the authors' contributions from the Acknowledgements and included them in a separate Authors' Contributions section. Also in that section, we have confirmed that all authors read and approved the final manuscript.
6. We have listed figure legends after the References.
7. We have included a completed CONSORT checklist as an additional file. Please note that this is not the first and main outcome paper of this trial.

Yours sincerely,

Prof Mark Tomlinson  
Contact email: markt@sun.ac.za